Contents lists available at ScienceDirect



Technological Forecasting & Social Change

journal homepage: www.elsevier.com/locate/techfore

The impact of public research on the technological development of industry in the green energy field



Technological Forecasting Social Change

Lorenzo Ardito^a, Antonio Messeni Petruzzelli^a, Claudia Ghisetti^{b,c,*}

^a Department of Mechanics, Mathematics, and Management, Politecnico di Bari, Viale Japigia 182/B, 70126 Bari, Italy

^b European Commission, Joint Research Centre (JRC), Via E. Fermi 2749, TP 361, Ispra, VA I-21027, Italy

^c Catholic University of Milan, Department of Economic Policy, Via Necchi 5, 20123 Milano, Italy

ARTICLE INFO

Keywords: Green energy technologies Public research Technology attributes R&D policy Environmental innovations

ABSTRACT

This paper, by analysing technology attributes, examines the degree to which companies build on environmental technologies generated by public research organisations (PROs) to develop the related technological solutions. Three main technology attributes have been considered: (i) the level of establishment, (ii) the scope of application, and (iii) the technological breadth. We have then developed three hypotheses about the influence exerted by these characteristics and have chosen the green energy as the field of investigation. The analysis is based on a sample of 4363 green energy patents registered by PROs at the United States Patent and Trademark Office in the 1976–2011 period. The results of a set of Tobit regression have revealed that the level of establishment, the scope of application and the technological breadth of public environmental technologies are positively related to the technological development of an industry. Hence, these findings can help advance the current debate on whether the green outputs of public research can stimulate firms to generate environmental technologies.

1. Introduction

The need to reduce human-induced climate changes, while avoiding shrinking economic growth is crucial to maintain sustainable economic growth (EC/EACI, 2011; OECD, 2012). In order to achieve this win-win effect, the development of technological solutions that favour the creation of and at the same time address environmental issues has become of foremost importance (Alkemade et al., 2011; Ghisetti et al., 2015; Mowery et al., 2010). Green energy technologies (i.e., alternative energy production and energy conservation solutions) are of utmost importance in this respect. On the one hand, since energy production and consumption account for two-thirds of the world's greenhouse-gas (GHG) emissions (OECD/IEA, 2015), "effective action in the energy sector is essential to tackle the climate change problem" (OECD/IEA, 2015: 20). On the other hand, greening the energy sector is necessary to create new business opportunities (OECD, 2012). The invention, adoption and diffusion of green energy technologies should therefore be at the core of a sustainability transition in order to replace the current dirtier technologies, improve environmental protection, and boost the economy (Benson and Magee, 2014; Geels et al., 2016; OECD/IEA, 2015).

However, market failures often hamper the capabilities of firms to capture the full economic return of their environmental R&D activities

(Mowery et al., 2010), which are subject to the so-called double externality problem (Jaffe et al., 2005; Rennings, 2000; Popp et al., 2010; Hoppmann et al., 2013). In other words, environmental R&D activities lead to two types of positive externalities, that is, knowledge creation and environmental benefits; nonetheless, neither are well captured by market prices (Rennings, 2000). As a result, only a suboptimal number of green energy technologies would be generated by the industrial sector in the absence of a proper regulatory push (e.g. Jaffe and Palmer, 1997; Johnstone et al., 2012), thus making governmental intervention essential to compensate for this misalignment (Foray et al., 2012; Ghisetti and Pontoni, 2015). The recent increase in public research organisation (PRO) spending on environmentally friendly energy solutions is seen as a means to compensate for such sub-optimal investment (Albino et al., 2014; Hargadon, 2010) and may be explained by both neoclassical and evolutionary economic theories. As far as the former are concerned, the public funding of PROs (e.g., universities, public research centres, governmental organisations) may compensate for market failures and generate new technological solutions that i) would not be developed by private research alone and ii) may be of use to as many firms as possible to generate related technological advancements. Instead, as far as the evolutionary economic theory is concerned, the availability of new technologies developed by PROs would not automatically lead to the setting of the basis for the development of

* Corresponding author at: European Commission, Joint Research Centre (JRC), Via E. Fermi 2749, TP 361, Ispra, VA I-21027, Italy. *E-mail addresses:* lorenzo.ardito@poliba.it (L. Ardito), antonio.messenipetruzzelli@poliba.it (A.M. Petruzzelli), claudia.ghisetti@unicatt.it (C. Ghisetti).

https://doi.org/10.1016/j.techfore.2019.04.007 Received 9 August 2018; Received in revised form 18 January 2019; Accepted 9 April 2019 Available online 17 April 2019 0040-1625/ © 2019 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/BY/4.0/). subsequent technologies by firms. In fact, firms are often not always willing and/or able to draw on public research outcomes because of their weaker connection to market needs or a lack of firms' absorptive capacity (Cohen and Levinthal, 1990; Salter and Martin, 2001). Consequently, the degree to which firms benefit from the research efforts of PROs has yet to be fully understood (Coccia and Rolfo, 2008; Cohen et al., 2002), and this knowledge can be used to improve the effectiveness of public investments in producing research externalities for the industrial sector (Beise and Stahl, 1999; Bornmann and Marx, 2014).

The present paper aims to be part of this debate and it deals with an application to the green energy field. In fact, its focus is not on the public incentives that may stimulate private environmental R&D (R&D tax credits, R&D subsidies, etc.), as is the case for most of the existing research (e.g., Hoppmann et al., 2013), but is instead on whether and under which conditions the knowledge underlying green energy technologies created by PROs spills to firms intensively and whether it boosts a firm's contribution towards technological development.

The empirical analysis has drawn on a sample of 4363 patents registered by PROs during the 1976–2011 period at the United States Patent and Trademark Office (USPTO) and has labelled them as environmentally based on their International Patent Classification (IPC) code. The results, obtained through Tobit regressions, have revealed that the level of establishment, the scope of application, and the technological breadth of public environmental technologies all have a positive effect on the degree to which the industrial sector builds on green public knowledge for further technological development.

The remainder of the paper is structured as follows. Section 2 presents the theory and hypotheses. Section 3 outlines the methodological approach, while Section 4 shows the results. Finally, Section 5 provides the discussion, implications, limitations and future research directions.

2. Theory and hypotheses

The paper assesses whether and to what degree companies build upon the technological knowledge generated by PROs, ultimately stimulating an industry to generate new (environmental) technologies. To the best of our knowledge, this is still an underdeveloped area of research. A recent exception is that of from Popp (2017), whose contribution has shed light on the value of knowledge flows between academia, the private sector and governmental research by means of US patent data which were used to answer questions about whether quality science has been useful for future technology development in the green energy field (namely biofuels, solar and wind) and about which institutions produce the most valuable research. His findings suggest the importance of public institutions in increasing the impact (citations) on renewable energy research and in helping new energy technologies overcome barriers to their exploitation (Mowery et al., 2010).

Our contribution adds to the above work by conducting a specific investigation on the nature of the attributes of the green energy technologies developed by PROs. Since the evolutionary economic theory recognises that not all the technologies developed by the public sector affect the industrial domain (Bornmann and Marx, 2014; Coccia and Rolfo, 2008; Cohen et al., 2002), this paper attempts to understand what types of PRO technologies lead to the technological development of industry. Technologies are not all equal and, depending on their attributes, they may be more (or less) attractive to and useful for firms (see, e.g., Chen et al., 2011; Shane, 2001). Such an investigation would allow more light to be shed on how the outputs of PRO research can help the transition to more environmentally friendly R&D efforts in the industrial sector, a transition which would require high levels of investment for the development of environmental technologies and thus riskier research activities that are unlikely to be sustained by the private sector on its own (Mowery et al., 2010), given the double externality features of such research activities (Rennings, 2000).

only may the technologies developed by PROs be attractive to different extents to firms, but they may also facilitate the transition of firms towards more environmentally friendly research activities, hence overcoming the double externality problem. Therefore, according to the previous discussion, the main research question of this study is: *how does the research generated by PROs impact the technological development of industry in the green energy field*? In other words, we question whether the attributes of the technological knowledge generated by PROs can explain if, and to what extent, such knowledge impacts the industry technological development of industry.

The technology attributes under investigation were selected on the basis of a review of the existing literature on technology management (e.g., Chen et al., 2011; Messeni Petruzzelli et al., 2015; Nerkar and Shane, 2007; Shane, 2001; Sohn et al., 2013) and of the emerging literature on environmental technology development (e.g., Popp, 2006; Popp et al., 2010; Nemet, 2012; Guan and Yan, 2016; for a review, see Barbieri et al., 2016).

By focusing on the role of knowledge flows from PROs as a determinant of a firm's technological development, the paper contributes to the current debate on the conditions under which the outcomes of public environmental research can stimulate the generation of more environmentally friendly technologies by industry, which is becoming more and more important to promote economic growth and address environmental issues (Alkemade et al., 2011). Furthermore, an alternative perspective is provided to evaluate the societal impact of public environmental research which, although difficult for governments to undertake, is necessary to better design effective funding policies (Bornmann and Marx, 2014).

2.1. Technology attributes

Hereafter, we argue that the attributes of the (environmental) technologies created by PROs can predict whether public research will set the basis for a firm's further technological development in the case of such technology-intensive industries as the energy sector. In fact, technology attributes reflect the opportunities and knowledge base that characterise technological solutions (Shane, 2001). These attributes may explain to what degree firms build upon certain types of technologies, since they choose technologies for subsequent development on the basis of their expectations for the future and their ability to exploit them (Nerkar and Shane, 2007; Shane, 2001). Technology attributes of PRO green energy technologies reflect the public environmental knowledge and opportunities available to firms that can be drawn on for subsequent technological development. Those attributes might thus alleviate/strengthen the double externality problem, mainly by fostering (or hampering) the intentions of firms to acquire the initial environmental knowledge needed for their research activities, but not internally developed by them, given the impossibility of capturing its full value.

After reviewing the extant literature on technology management and environmental technology drivers, three attributes have been identified, namely (i) the level of establishment (Heeley and Jacobson, 2008), (ii) the scope of application (Novelli, 2015) and (iii) the technological breadth (Messeni Petruzzelli et al., 2015). The level of establishment reflects to what extent technologies are known and their underlying knowledge has been exploited; the more (/less) known and exploited the technologies are, the more established (/nascent) they can be considered (Ahuja and Lampert, 2001; Heeley and Jacobson, 2008). The scope of application refers to the extent of different domains a technology is related to, i.e., a higher scope reflects a broader set of potential applications (Gambardella et al., 2007; Lerner, 1994). Finally, technological breadth captures the extent to which the knowledge base of a technology is diversified, thus reflecting a wider search process during its development (Messeni Petruzzelli et al., 2015).

It will be argued that, depending on the technology attributes, not

From a technology management perspective, it has been proved that such attributes may point out if and how a firm relies on related (public) technologies, for instance by explaining the likelihood of firm formation (Shane, 2001), technology selling/acquisition (Nerkar and Shane, 2007; Sohn et al., 2013) and the commercialisation of environmental technologies (Chen et al., 2011). Furthermore, some studies have revealed that the three selected attributes affect the technological impact of patents (e.g., Fischer and Leidinger, 2014; Messeni Petruzzelli et al., 2015; Sterzi, 2013) - i.e., the impact of a patent on the development of any related and subsequent patents (e.g., Ahuja and Lampert, 2001) - even in the case of green energy technologies (Nemet, 2012).

As far as the setting under examination is concerned, the development of green energy technologies involves technological knowledge that covers a long time period and/or is subject to a faster (or slower) exploitation process (Nemet, 2012), hence highlighting the need to investigate the level of establishment of PRO technologies. Furthermore, the green energy sector may affect multiple industries (e.g., transportation, waste management, and energy production) and the respective technological development may rely on multiple knowledge domains (Benson and Magee, 2014; Guan and Yan, 2016), given the complexity that characterises such solutions (Carrillo-Hermosilla et al., 2010), which are usually more distant from the traditional sectorial knowledge base (De Marchi, 2012). Accordingly, Popp and Newell (2012) outlined that alternative energy technologies are more general than dirtier technologies, serve multiple sectors and are hence cited more often than other technologies. Furthermore, clean energy innovations are found to create more knowledge spillovers than dirtier innovations (Dechezleprêtre et al., 2014), so that their inventions are more pervasive and applicable than dirtier ones. It has recently been confirmed that environmental technologies differ significantly from non-environmental ones under multiple attributes (e.g., originality, radicalness and generality), including their scope (Barbieri et al., 2018). As far as the latter aspect is concerned, the higher a patent scope is, the greater the complexity of an invention, the higher the number knowledge components it would draw upon (Lerner, 1994). Green technologies have in fact been found to draw on more dispersed technological fields and knowledge components than other technologies (Barbieri et al., 2018).

Overall, this evidence would seem to suggest the need to investigate the energy technology realm to assess the effect of the selected technology attributes on the generation of green knowledge, given the specificities and differences from standard technologies.

2.1.1. The level of establishment

Prior research suggests that established technologies may positively impact subsequent technological development, since they are more codified than novel ones, and hence require an lower level of absorptive capacity (Dosi and Nelson, 2010; Heeley and Jacobson, 2008). Thus, it is easier and more cost-effective for firms that have not developed solutions to build on established technologies with fewer risks. Second, a better understanding of a given technology provides firms with more chances to come up with useful recombinant opportunities, which may then act as a catalyst for the generation of new technologies (Fleming, 2001; Savino et al., 2015). Third, the exploitation of established technologies is legitimated more by the operating market, because customers usually consider them as more familiar (Story et al., 2014). The evidence that green energy technologies that are more radical (especially in a systemic vision) and at their initial stage are considered by firms as less reliable and/or more costly than dirtier and already established solutions (Oltra and Saint Jean, 2009) is particularly relevant for the energy sector. This perception, in turn, increases the rigidity of the energy regime, which makes the entry and switching costs of less established green energy technologies higher (Geels et al., 2016; Wainstein and Bumpus, 2016), so that, in the absence of policy stimulus (including PRO innovations), technological lock-in and path dependence may lead firms and the market to favour established (and dirtier) solutions for future economic activities (Oltra and Saint Jean, 2009).

Thus, the argument is that the higher the level of establishment of a PRO technology is, the more likely it is to influence a firm's own R&D efforts in the green energy field.

According to evidence collected for "standard" technological solutions, this positive effect deteriorates after a certain threshold, as a technology loses its potential exploitability (U-shaped effect) (e.g., Capaldo et al., 2017; Ardito et al., 2016a). Notwithstanding this, we do not expect to experience any inverted U-shaped effect in the green energy realm. In fact, we argue that the energy sector faces certain specificities that make the level of establishment of PRO technologies positively and linearly supportive of the technology development of an industry. These specificities are the high path dependency of the energy sector (see, e.g., Oltra and Saint Jean, 2009) and the evidence that industrial economies have suffered from a lock-in effect pertaining to "fossil fuel-based energy systems", which were augmented by the coevolution of technological and institutional settings that were path dependent, to the advantage of dirtier technologies (Unruh, 2000). This is what has happened, for instance, in the automotive sector, in which the persistent dominant design of the more conventional engine technology (internal combustion regime) has led to continuous and incremental improvements to this technological path in order to make vehicles greener (more efficient and sustainable), at the expense of the less established technological path of alternative engine technologies (such as electric batteries, fuel cells, or hybrid vehicles) (Oltra and Saint Jean, 2009). Consequently, the transition to greener energy technologies is less mature and needs longer, so that no detrimental effect of the level of establishment may be expected. Therefore, PROs that aim at providing research externalities to industry may have a greater chance of attracting several companies, if they continue with the initial improvement of a certain green energy technology, instead of providing completely new solutions. In this way, they help firms rely on green energy solutions that might be relevant for the future objectives of the firms, but without the necessity of making them in-house, thus limiting the influence that the double externality problem has on the engagement of firms in environmentally friendly R&D activities (Hoppmann et al., 2013). Therefore, we expect that:

H1. The level of establishment of public environmental technologies has a positive effect on the related technological development of industry.

2.1.2. The scope of application

Recent research has revealed that non-innovating companies are more likely to build on technologies with a broad scope of application than on innovating ones (Novelli, 2015). Innovative firms in fact usually fail to protect and exploit broad technologies, because they do not have the complementary assets or internal capabilities required to pursue developments across all the potential domains in a timely manner (Gambardella and Giarratana, 2013; Rosenberg and Trajtenberg, 2004). On the other hand, given the high visibility of broad technologies, many non-innovating companies may come across them during their respective search processes and focus on the domain in which they are more expert or which needs a specific contribution to solve an ongoing issue (Lee and Lee, 2010; Novelli, 2015), and hence develop follow-up inventions that substitute the original ones. This in turn implies that the broader the technological solutions developed by PROs are, the more likely firms will be to come across those solutions and build on them. In other words, the scope of a technology signals its pervasiveness, and the higher this pervasiveness is, the more likely it is that relevant innovations will be developed, as occurs in the case of General Purpose Technologies (GPTs) (Bresnahan and Trajtenberg, 1995). The literature on GPTs shows interesting patterns that can partially be extended to this context, and which support the need to analyse technology attributes in depth. GPTs are currently benefiting from the recombination of knowledge from diverse knowledge areas, and their development is being affected by the technological attributes of the knowledge they build upon (Appio et al., 2017). The analysis of a patent's attributes (of any technology) may help to discern and predict whether it could become a transformative GPT or not (Feldman and Yoon, 2011). Technology attributes themselves affect the likelihood of their sustaining (or not) the diffusion of GPTs and the effects of GPTs on subsequent long-run economic growth (Andergassen et al., 2017).

The role played by this pervasiveness may be accentuated in the (green) energy sector because it is related to multiple domains (e.g., building, transportation, waste management and energy production) (OECD, 2012). Therefore, a number of different companies can encounter a green energy technology with a broad application scope.

Furthermore, the adoption of broad PRO technologies with environmental benefits may increase the competitiveness of firms. These solutions may be sources of competitive advantage for firms who build on existing technologies to develop new ones that expand existing (environmental) knowledge bases (De Marchi, 2012) without the costs associated with their internal development and, hence, going beyond a firm's core competencies (Teece et al., 1997). Moreover, even though several companies build on the same broad environmental technology, it may still be a source of competitive advantage, since not all the companies will adopt the technology on the same market or for the same application, given its pervasive nature. This is the case, for instance, of LED lighting technologies, which are considered important solutions within the energy conservation umbrella (WIPO, 2012). In fact, these technologies have been adopted by several firms in different kinds of industries (e.g., illumination, television, computers, and car manufacturing) to develop more energy efficient products, whose launching would have been hampered (or prevented) without the R&D programmes of some PROs, such as "LED lighting facts" by the US Department of Energy.¹ The foregoing discussion supports the hypothesis that:

H2. The scope of application of public environmental technologies has a positive effect on the related technological development of industry.

2.1.3. The technological breadth

Technologies resulting from wide searches are more likely to be understood by companies operating in different industries (Banerjee and Cole, 2010). This indicates that more companies may build on solutions with higher technological breadth. Accordingly, environmental technologies, such as energy ones, are multi-faceted in nature (Markard et al., 2012) and are more systemic than standard innovations (Carrillo-Hermosilla et al., 2010). Consequently, they may serve to spur further technological advancements. Moreover, the various companies that attempt to build on technologies with a high technological breadth may benefit from a large pool of potential links and associations between various knowledge domains, and this may in turn lead to more opportunities for knowledge recombination (Maggitti et al., 2013) and, hence, more chances of contributing to technological development (Fleming, 2001; Messeni Petruzzelli et al., 2015). Such recombinant opportunities are particularly relevant to allow firms to build on green energy technologies. Green energy technologies, based on multiple knowledge domains, are more likely to set the basis for future technological developments (Benson and Magee, 2014; Nemet, 2012) and thus they have a higher impact as they may be used to solve diverse technological problems while addressing environmental concerns (Ardito et al., 2016b). The work of Noailly and Shestalova (2017), who pointed out that renewable energy patents in general broadly serve external knowledge development, is an interesting example, as onethird of their citations came from technology fields other than power generation. More specifically, the authors established what external technology fields build most on the knowledge developed within the renewable energy field and found interesting specificities: solar patents

are mainly cited for semiconductors, thermal processes and apparatus, and civil engineering; *wind* patents for electrical machinery, engines, pumps and turbines, mechanical elements and transport; *storage* patents for electrical machinery; *waste* and *biomass* for basic material chemistry, chemical engineering and environmental technology patents.

It may be expected that those PROs that combine more knowledge domains during their environmental research processes will find solutions of a higher impact on the technological development of industry. In addition, in the same way as for the development of environmental technologies, these benefits have been observed from an open innovation mode. In fact, the breadth of external knowledge sourcing has been found to positively influence the rate of adoption of those solutions (Ghisetti et al., 2015), since relying broadly on multiple knowledge sources provides firms with valuable knowledge flows which are later transformed into actual environmental technologies. This is especially true for green energy technologies, for which the lack of external knowledge sourcing in R&D has been found to be a detrimental barrier to their subsequent adoption and exploitation (Rennings and Rammer, 2009). Therefore, we hypothesise that:

H3. The technological breadth of public technologies has a positive effect on the related technological development of industry.

3. Data and methodology

3.1. Industry setting

The green energy field is a suitable field to test the proposed hypotheses for many reasons. First, although PROs are usually set up to provide scientific knowledge, their roles as technology providers has gained increasing importance in this specific context (Albino et al., 2014; Oltra et al., 2012). Second, patents are widely used to identify and examine green energy technologies (Guan and Yan, 2016; Nemet, 2012; Popp, 2017; Veugelers, 2012). Moreover, patent data allow several attributes of patented inventions to be identified and the degree to which they have impacted subsequent technological developments to be established (Kaplan and Vakili, 2015; Messeni Petruzzelli et al., 2015; Nerkar and Shane, 2007). Patent citations have also been discussed to obtain a good indication of knowledge transfer among inventors (Jaffe et al., 2000). Finally, many governments have changed their intellectual property policies to increase incentives for public research patenting (e.g., the Bayh-Dole Act). Consequently, the rate of PRO patenting has increased (Sterzi, 2013), especially in the green energy field (Albino et al., 2014; Hargadon, 2010). Therefore, patents can be considered as a reliable information source to conduct the present research.

3.2. Data collection

We first detected technologies developed by PROs in the green energy field by looking into patents registered at the USPTO and by labelling them as environmental on the basis of the IPC codes pertaining to the "Alternative energy production" and "Energy conservation" technology fields, as defined in the IPC Green Inventory (WIPO, 2012). All the patents related to the following macro categories: Bio-fuels; Fuel cells; Harnessing energy from manmade waste; Hydro energy; Wind energy; Solar energy; Geothermal energy; Other productions or uses of heat, not derived from combustion, e.g. natural heat and Using waste heat all belong to the first category. Those patents related to the following macro categories: Storage of electrical energy; Power supply circuitry; Low energy lighting; Thermal building insulation, in general and Recovering mechanical energy all belong to the second category. All the patents that matched our search criteria, but were not registered by a research organisation or were applied for after 2011, were excluded to allow enough time for the patents to be cited (e.g. Katila, 2002). Finally, websites, reports and other additional sources were

¹ See http://www.lightingfacts.com/About.

scrutinised to distinguish between patents registered by private and public research organisations. This procedure yielded a final sample of 4363 patents that had only been developed by PROs during the 1976–2011 period. The final sample included patents by PROs established in diverse countries, such as the US Department of Energy, the French Commissariat à l'énergie atomique et aux énergies, the National Research Council of Canada, the Indian Council of Scientific and Industrial Research, and the Taiwanese National Tsing Hua University, which are also among the most patent-intensive public organisations in the world.

It is important to stress that our empirical design selected the patents, developed by PROs, which had been taxonomised as belonging to the green energy field in existing classifications as belonging to the green energy field, as detailed before, and not those firms or, more generally, the population of actors belonging to the energy sector. In other words, our definition of the green energy sector was driven by the patents (which constitute our unit of analysis) and not by the actors active in the energy sectors. This implies that, on the one hand, it has not been possible to perfectly overlap the evidence obtained from looking at aforementioned patents with the evidence regarding the status quo and the overall evolution of the sustainability of the energy sector, which is discussed elsewhere, for instance in recent ad hoc reports (EEA, 2018). On the other hand, we have not restricted our analysis on the subsequent technological development of the sector to actors who necessarily belong to this sector, i.e. we have looked at the industrial use of those patents, generated by PROs, that have been labelled as being related to the green energy field, regardless of whether the user of those patents is connected to the energy sector or not.

3.3. Variables

3.3.1. Dependent variable

Forward citations have been widely considered as a proxy for the impact of patents on subsequent technological developments (Arts et al., 2013; Kaplan and Vakili, 2015; Messeni Petruzzelli et al., 2015). In fact, "external citations indicate that other players have internalised part of the knowledge underlying the original invention and succeeded in building on it" (Novelli, 2015: 494). In line with this reasoning, the citations received by the patents in the sample were analysed to assess whether they had been referenced by patents developed by companies, in order to test whether PRO patents have had an impact on the technological development of the related industry. In other words, we identified the citing patents for each patent in our sample and to whom they belong. Once the owners of the citing patents had been identified, a distinction was made between those registered by firms and those registered by other types of assignees, such as universities and research organisations, single inventors and financing institutions. Finally, after checking the company names, the dependent variable (In*dustryTechDevelopment*) was built as follows:

IndustryTechDevelopment =
$$1 - \sum_{F} \left(\frac{C_{FP}}{C_{P}}\right)^{2}$$
,

where C_{FP} indicates the number of times firm F cited focal patent P, while C_P is the total number of citations of patent P from patents owned by firms. This is a Herfindahl-type index, which is based on the generality index proposed by Trajtenberg et al. (1997). Unlike Trajtenberg et al.'s (1997) generality index, this measure does not account for the diverse technological classes of the citing patents; it instead considers the diverse firms that own the cited patents. The rationale behind this index is that the greater the variety of firms that own the cited patents is, the higher the impact of the PRO patent on the technological development of the industry.

This variable may alternatively be computed as a simple count of diverse firms that own at least one patent which cites a PRO patent. However, in this way, the actual degree to which the PRO patent affects the industrial sector would not emerge. This can be understood by considering the following example: let a patent receive 15 citations equally distributed among patents owned by companies A, B and C, and let a second patent also receive 15 citations, but 13 of them are from patents owned by company A, one citation is from company B, and one citation is from company C. The first invention actually has a wider impact on the industry than the second one, which mainly has an impact on company A. In addition, if we adopted a count measure, its value would be the same for the two patents (three). On the other hand, the Herfindahl index, *IndustryTechDevelopment*, would have amounted to 0.88 for the first invention and 0.24 for the second invention. The latter would better characterise the diverse impact of the two inventions, but the former would reflect a wider impact.

Furthermore, although this study covers a long period of time and there may be a bias towards older patents, in terms of forward citations, we are confident that our measure is still reliable for the following reasons. First, the most recent patents in our sample were filed for in the year 2011, with forward citations collected until the beginning of 2017. Hence, we have captured all the received citations for at least five years after the patent applications, which is considered the time span when patents obtain most of their citations (e.g., Griliches, 1987; Katila, 2002; Nooteboom et al., 2007). Second, our models include time-period effects to account for systematic intertemporal differences (see Section 3.3.3).

3.3.2. Independent variables

The first independent variable measures the extent to which a technology is established (Established). It is computed as the natural logarithm of the number of a patent's backward citations. The logarithmic transformation was adopted to correct for skewness and kurtosis.² The rationale behind this measure is based on previous studies that argued that the higher the citations made by a focal patent are, the higher the level of establishment (e.g., Ahuja and Lampert, 2001; Ardito et al., 2016a; Nerkar and Shane, 2007; Ziedonis, 2007). In fact, since a technology is often the result of a cumulative process over a technological trajectory (Dosi, 1982; Dosi and Nelson, 2010), it can be assumed that patents that refer to a large number of previous patents reflect inventions with a stronger link to a well-established technological paradigm (Dornbusch and Neuhäusler, 2015; Martinelli, 2012). Conversely, a small number of backward citations signal that a patent is disconnected from previous technical solutions, thus highlighting its novel nature (Ahuja and Lampert, 2001; Ziedonis, 2007). Second, according to the literature in this field, the scope of application (Scope) was operationalised as the number of different three-digit US classes assigned to each patent by USPTO (Lerner, 1994; Novelli, 2015). Finally, in order to assess the breadth of the technological base upon which a patent was built (TechBreadth), Trajtenberg et al.'s (1997) originality index was adopted (see also Arts et al., 2013). This index is computed through a Herfindahl index, which highlights the degree of diversification, in terms of technological classes, of the patents upon which a focal patent is based (i.e. the backward citations). Specifically:

Technological breadth =
$$1 - \sum S_{ij}^2$$
,

where S_{ij} refers to the fraction of patents cited for patent i that belong to the three-digit US class j out of n technological categories assigned to the patents by USPTO.

3.3.3. Control variables

Several control variables were included to improve the reliability of our analyses. First, the scientific nature of the knowledge underlying a patented technology (*Scientific*) was accounted for by counting the

 $^{^{2}}$ The skewness and kurtosis values before the logarithmic transformation were 7.78 and 93.81, respectively. After the logarithmic transformation, they decreased to zero and three, respectively.

number of patent references to non-patent documents (Narin et al., 1997). Since this variable was skewed and kurtotic, a logarithmic transformation was also used.³ Second, a dummy variable that assessed whether a patent had been jointly developed by several organisations was included (value one) (Joint) (Messeni Petruzzelli et al., 2015). Third, we controlled for the number of patent claims (Claims) (Novelli, 2015; Reitzig, 2004). Fourth, we accounted for the number of inventors (Singh, 2008) who had been involved in the invention process (Team-Size). Fifth, since a patent may attract the attention of companies, whether it has been published in other patent offices or not, and whether the innovating organisation has been willing to pay renewal fees or not, we took into account the number of patent offices (besides USPTO) where the patent had been registered (PatFamily) (Reitzig. 2004) and whether the patent renewal fees had been paid at least once (Renewal) (Fischer and Leidinger, 2014). Sixth, we distinguished between PROs from universities, governmental organisations (e.g., the U.S. Department of Energy) and other public research centres (e.g., national research councils), each of which was represented by a dummy variable. Thus, we included two dummies (dummy PRO) to control for the different types of PRO, with the "University" PRO being the benchmark. Seventh, we also included dummy variables to control for relevant time period effects (dummy period). Time fixed effects were considered for 1976-1987, 1988-1997, 1998-2002 and 2003-2011, and the latter was considered as the benchmark. Eighth, we added a dummy variable to assess whether a PRO that owned a patent was US based (value one) in order to control for a potential country bias (dummy US). Finally, we included a dummy variable that took on the value of one if a patent was an "Alternative energy production" technology (dummy AEP), and zero in the case it referred to an "Energy conservation" technology.

3.4. Model specification

Since our focus has been on the technology attributes of an invention, the unit of analysis of this study is a single patent. Specifically, *IndustryTechDevelopment* has been operationalised through a Herfindahl index, which takes on continuous values, ranging from zero to one. Given the double bounded nature of this variable and its non-normal distribution, it falls into the category of limited dependent variables (LDVs) (Long, 1997). Therefore, we employed a Tobit regression (see, for example, Banerjee and Cole, 2010; Köhler et al., 2012). This type of regression is in fact suitable when the dependent variable is limited to a range of values and is not normally distributed (Long, 1997). Conversely, other types of regressions (e.g., OLS) cannot account for the nature of LDVs and tend to estimate incorrect parameters (Long, 1997; Wiersema and Bowen, 2009; Wooldridge, 2012). Furthermore, the bias in the fixed effect estimates, due to the "incidental parameter problem", is attenuated in the case of Tobit models (Greene, 2004).

4. Results

Tables 1 and 2 show the descriptive statistics and pairwise correlations, respectively. According to Table 2, the correlations are essentially low, with the exception of 0.66 for Technological Breadth and Establishment, but the maximum variance inflation factor for each model is below two, hence signalling the absence of severe multicollinearity issues (Cohen et al., 2013).

Table 3 presents the results of the Tobit regression. Model 1 is the baseline model, which only includes the control variables. Models 2–4 are the partial models, each of which contains one of the three independent variables. Model 5 is the full model. Model 1 reveals that

PRO green energy patents have a wider impact on the industrial sector if the number of claims is high ($\beta = 0.008$, p < 0.01), and if they are owned by several organisations ($\beta = 0.057$, p < 0.10), have been renewed at least once ($\beta = 0.244$, p < 0.01) and the PRO is US based (β = 0.110, p < 0.01). The opposite effect can be ascribed to the reliance on basic research ($\beta = -0.081$, p < 0.01), team size $(\beta = -0.011, p < 0.05)$, and the development of an "Alternative energy production" technology ($\beta = -0.246$, p < 0.01). Model 2 tests H1 and suggests that the level of establishment is positively related to IndustryTechDevelopment ($\beta = 0.114$, p < 0.01), according to H1. In line with H2, Model 3 reveals that the scope of application has a positive impact on the technological development of an industry $(\beta = 0.027, p < 0.01)$. Finally, Model 4 supports H3, in that the coefficient of *TechBreadth* is positive and significant ($\beta = 0.296$, p < 0.01). We added the squared term of the independent variable under analysis to establish robustness for each partial model. No squared terms resulted significant, thus confirming the initial hypotheses (see Models 2a, 3a, and 4a). The full model also corroborated the results of the partial models. In addition, we measured the patent scope considering the number of diverse sub-classes instead of the number of different three-digit IPC classes. All the hypotheses continued to be supported in this case.

5. Discussion and conclusion

Boosting the development of green energy solutions by the private sector is pivotal for policymakers in order to simultaneously improve economic growth and environmental benefits (Hoppmann et al., 2013; Rennings, 2000). The aim of public research is to correct market failures (e.g., the double externality problem) while hampering private environmental R&D efforts, yet PROs often fail to achieve this goal (Balachandra et al., 2010). Furthermore, it has been pointed out that green technologies differ substantially from non-environmental ones, both as far as their nature and their impacts are concerned, not only because of their "double externality" nature, but also because they are more complex and have a greater and more pervasive impact on the subsequent related inventions (Barbieri et al., 2018).

Therefore, the present paper has been aimed at understanding under which conditions the research outcomes of PROs influence the technological development of an industry in the green energy sector. To do so, the attributes of PRO technologies (i.e., the level of establishment, the scope of application, the scientific nature, and the technological breadth) have been analysed.

A Tobit regression, which was conducted on a sample of 4363 patents registered by PROs during the 1976–2011 period, and which were classified as "Alternative energy production" or "Energy conservation" technology fields, revealed that the level of establishment, the scope of application and the technological breadth had a positive effect on the degree to which many companies will build on PRO research outcomes, thus supporting our conjectures. Considering that "understanding the impact of public research on industrial R&D is central to understanding the innovation process itself" (Cohen et al., 2002: 1), these results provide relevant theoretical and policy implications.

5.1. Theoretical implications

From a theoretical perspective, this research has investigated a novel and understudied set of antecedents (i.e., technology attributes) that may help explain when the outcomes of public R&D efforts in the green energy field provide externalities for the industry sector. To the best of our knowledge, prior research has widely investigated the policy incentives that foster private R&D activities (e.g., public subsidies) (e.g., Hoppmann et al., 2013; OECD, 2012), but negligible efforts have been devoted to examining situations in which the results of public research have an impact on the industrial sector by explicitly focusing on the same nature of the technological results as this research.

 $^{^3}$ The skewness and kurtosis values before the logarithmic transformation were 11.57 and 254.98, respectively. After the logarithmic transformation, they decreased to 0.68 and 2.72, respectively.

Table 1Descriptive statistics.

	Mean	Std.	Min	Max	Description
IndustryTechDevelopment	0.287	0.350	0.000	0.972	Modified generality index (Trajtenberg et al., 1997), with a focus on the owners of citing patents instead of their technological classes.
Established	1.759	0.966	0.000	5.505	# backward citations.
Scope	2.043	1.100	1.000	10.00	# different three-digit US classes.
TechBreadth	0.3861	0.299	0.000	0.9158	Originality index (Trajtenberg et al., 1997).
Scientific	1.434	1.375	0.000	6.982	# citations to non-patent documents.
Joint	0.121	0.327	0.000	1.000	1 if a patent is owned by more organisations, 0 otherwise.
Claims	14.793	11.337	1.000	99.00	# claims.
TeamSize	2.971	1.938	1.000	24.00	# patent inventors.
PatFamily	1.391	1.333	0.000	5.000	# patent offices, besides the UPTO, in which the patent is also published.
Renewal	0.678	0.467	0.000	1.000	1 if the patent has been renewed, 0 otherwise.
Dummy US	0.540	0.498	0.000	1.000	1 if a patent is owned by a US PRO, 0 otherwise.
Dummy AEP	0.786	0.410	0.000	1.000	1 if a patent is in the "Alternative energy production" technology field, 0 otherwise.

n = 4363.

Pairwise correlations.

	1	2	3	4	5	6	7	8	9	10	11	12
1-IndustryTechDevelopment	1											
2-Established	0.119*	1										
3-Scope	0.067*	-0.006	1									
4-TechBreadth	0.113*	0.666*	0.150*	1								
5-Scientific	-0.239*	0.047*	0.148*	-0.011	1							
6-Joint	-0.084*	-0.002	0.078*	-0.017	0.207*	1						
7-Claims	0.075*	0.169*	0.126*	0.141*	0.221*	0.020	1					
8-TeamSize	-0.163*	-0.024	-0.011	-0.022	0.123*	0.206*	0.002	1				
9-PatFamily	-0.141^{*}	-0.035^{*}	0.133*	-0.032^{*}	0.314*	0.169*	0.156*	0.119*	1			
10-Renewal	0.129*	0.027	0.027	0.007	0.057*	0.010	0.120*	0.024	-0.013	1		
11-Dummy US	0.209*	0.099*	0.019	0.040*	0.125*	-0.100^{*}	0.077*	-0.181^{*}	-0.254*	0.035*	1	
12-Dummy AEP	-0.053*	-0.176^{*}	0.104*	-0.202^{*}	0.145*	0.127*	-0.087^{*}	0.014	0.090*	-0.057*	0.116*	1

n = 4363.

* p < 0.05.

We have revealed that investigating the role of technology attributes is relevant to assess the impact of public environmental research on the technological development of an industry, in particular with reference to the green energy field, which has been at the core of academic (e.g., Barbieri et al., 2016; Hoppmann et al., 2013) and political debates (OECD, 2012; OECD/IEA, 2015). These technologies present distinctive features with respect to standard technologies, and hence call for ad hoc investigations. Moreover, their role has been proved to be central to move towards a low-carbon society (Benson and Magee, 2014; Nemet, 2012). Notwithstanding this, these studies have not provided insights about the potential influence of technology-level factors, such as technological attributes, to explain whether green energy technologies, developed by PROs, have an impact on the subsequent research and technological activities of firms.

Moreover, among the few studies that have examined the results of public research in promoting R&D initiatives related to the private sector (e.g., university-industry-government collaborations and Spin-off creation) (e.g., Bruneel et al., 2010; Dornbusch and Neuhäusler, 2015), very few of them have narrowed the level of analysis to the technology level (e.g., Shane, 2001), which, instead, has been proved to be pivotal to understand innovation phenomena and the impact of developed technologies (e.g., Chen et al., 2011; Nerkar and Shane, 2007). In the same way, studies that have examined public-private knowledge transfer have mainly focused on firm/university characteristics, modes of knowledge transfer and proximity dimensions (e.g., geographic) (Agrawal, 2001; Azagra-Caro et al., 2017; Lindelöf and Löfsten, 2004). However, except for in the work of Bekkers and Bodas Freitas (2008), knowledge specific characteristics (e.g., codification) have been undervalued; additionally, whether these knowledge characteristics help firms improve their innovation processes has remained unclear. These arguments help to point out the relevance of our contribution, which

refines earlier works on the influence of public research on the industrial sector by offering empirical evidence about the role of technology attributes.

Finally, this study also contributes to the discussion on "where" to measure the influence of a (green) technology. In fact, a technology may influence the technological landscape in which R&D efforts succeed within or beyond a technological, geographic or institutional domain (see, for example, Ardito et al., 2016b; Messeni Petruzzelli et al., 2015). In line with this perspective, our results highlight when (green) technologies belonging to the public domain have an impact on the industrial sector. These results may be added to the literature on research evaluation (e.g., Bornmann and Marx, 2014; Lyall et al., 2004). This literature claims that measuring the societal benefits of public research is still a complex task, and we believe our approach may have provided further insights on how to measure the societal impact of public research, in light of its influence on achieving sustainable goals.

5.2. Policy implications

From a policy perspective, we have provided guidance on how governments can allocate public funds to develop green energy technologies in order to play a relevant role in the technological development of an industry. We advise policymakers that the attractiveness of such public technologies is not the same for all companies. Therefore, when it is relevant to produce research externalities for the industrial sector, policymakers should design funding schemes that are devoted specifically to developing explicit types of technical solutions. In turn, it is important that PROs receive clear indications regarding the R&D outcomes they are expected to produce. Therefore, we suggest that policymakers should invest in strengthening their networks with corporate executives in the green energy field to obtain a better

$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	add 0.114*- 0.010 0.114*- 0.010 0.114*- 0.010 add -0.008 0.009 0.007 0.007 0.007 0.007 0.003 add -0.008 0.009 -0.008* 0.009 -0.008* 0.003 0.014 add -0.008* 0.009 -0.008* 0.003 -0.008* 0.003 -0.001 0.007* 0.003 0.007* 0.003 0.007* 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 0.003 -0.001 -0.001 0.003 -0.014 0.003 -0.014 0.003 -0.014 0.003 -0.014 0.003 -0.014 0.003 -0.014 0.003 -0.014 0.003 -0.014 0.003 -0.014 -0.003 -0.014 <td< th=""><th>media 0.114** 0.010 0.114** 0.010 0.014** 0.010 media -0.008 0.009 0.007** 0.007** 0.007** 0.037 0.037 0.037 adth -0.008*** 0.009 0.007** 0.009 0.007** 0.009 0.007** 0.001 0.007** 0.014 0.001<</th><th></th><th>Model 1</th><th>s.e.</th><th>Model 2</th><th>s.e.</th><th>Model 2a</th><th>s.e.</th><th>Model 3</th><th>s.e.</th><th>Model 3a</th><th>s.e.</th><th>Model 4</th><th>s.e.</th><th>Model 4a</th><th>s.e.</th><th>Model 5</th><th>s.e.</th></td<>	media 0.114** 0.010 0.114** 0.010 0.014** 0.010 media -0.008 0.009 0.007** 0.007** 0.007** 0.037 0.037 0.037 adth -0.008*** 0.009 0.007** 0.009 0.007** 0.009 0.007** 0.001 0.007** 0.014 0.001<		Model 1	s.e.	Model 2	s.e.	Model 2a	s.e.	Model 3	s.e.	Model 3a	s.e.	Model 4	s.e.	Model 4a	s.e.	Model 5	s.e.	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	th 0.027* 0.037 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.037 0.036 0.031 0.037 0.036 0.037 0.036 0.031 0.037 0.036 0.031 0.036 0.031 0.036 0.031 0.036 0.031 0.036 0.031 0.036 0.031 0.036 0.031 0.032 0.031 0.036 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.031 0.032 0.032 0.031 0.032 0.032 0.031 0.032 0.032 0.032 0.031 0.032 0.032 0.032 0.032 0.032 0.032 0.032 0.032 0.032 0.032 0	0.027+ 0.009 0.067* 0.027 0.003 0.035 0.035 0.035 0.036 0.037 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 0.036 <	Established Fstablished 2			0.114***	0.010	0.114*** -0.008	0.010									0.095***	0.013	
adth adth2 adth	dth 0.099 -0.087* 0.098 -0.087* 0.009 -0.087* 0.009 -0.081* 0.009 -0.087* 0.009 -0.081* 0.009 -0.081* 0.009 -0.081* 0.009 -0.017* 0.009 -0.081* 0.009 -0.011* 0.009 -0.011* 0.0011 <th colsp<="" td=""><td>dth 0.296*** 0.009 -0.081** 0.009 -0.081** 0.003 0.003*** 0.003*** 0.003*** 0.003*** 0.003*** 0.003**** 0.003**** 0.003**** 0.003***** 0.003**********************************</td><td>Scope Scone2</td><td></td><td></td><td></td><td></td><td></td><td></td><td>0.027</td><td>0.009</td><td>0.067** - 0.007</td><td>0.027 0.005</td><td></td><td></td><td></td><td></td><td>0.020**</td><td>0.00</td></th>	<td>dth 0.296*** 0.009 -0.081** 0.009 -0.081** 0.003 0.003*** 0.003*** 0.003*** 0.003*** 0.003*** 0.003**** 0.003**** 0.003**** 0.003***** 0.003**********************************</td> <td>Scope Scone2</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>0.027</td> <td>0.009</td> <td>0.067** - 0.007</td> <td>0.027 0.005</td> <td></td> <td></td> <td></td> <td></td> <td>0.020**</td> <td>0.00</td>	dth 0.296*** 0.009 -0.081** 0.009 -0.081** 0.003 0.003*** 0.003*** 0.003*** 0.003*** 0.003*** 0.003**** 0.003**** 0.003**** 0.003***** 0.003**********************************	Scope Scone2							0.027	0.009	0.067** - 0.007	0.027 0.005					0.020**	0.00
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	-0.082* 0.009 -0.081* 0.009 -0.085* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.009 -0.079* 0.003 0.001 0.007*	-0.082* 0.009 -0.081* 0.009 -0.079* 0.009 -0.011* 0.005 0.001* 0.009 -0.079* 0.001 0.009 -0.011* 0.005 0.001* 0.003 0.001* 0.003 0.011* 0.005 0.001* 0.003 0.011* 0.005 0.011* 0.005	TechBreadth TechBreadth2											0.296***	0.032	0.355*** - 0.081	0.108	0.096**	0.041	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.057 ⁺ 0.032 0.047 0.032 0.047 0.032 0.047 0.032 0.047 0.032 0.047 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.041 0.003 0.001 0.006* 0.001 0.005* 0.0011 0.003 0.0011* 0.003 0.0011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011* 0.003 0.011*	0057 ⁺ 0.032 0.047 0.032 0.047 0.032 0.047 0.032 0.047 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.049 0.032 0.011 0.003 0.001 0.003 <	Scientific	-0.082	0.009	-0.081***	0.009	-0.079	0.009	-0.084***	0.009	-0.085***	0.009	-0.080****	0.009	-0.079	0.009	-0.082	0.009	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.008** 0.001 0.007** 0.001 0.007** 0.001 0.007** 0.001 0.007** 0.001 0.007** 0.001 0.007** 0.001 0.006*** -0.011* 0.005 -0.011* <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td>Joint</td> <td>0.057^{+}</td> <td>0.032</td> <td>0.047</td> <td>0.032</td> <td>0.047</td> <td>0.032</td> <td>0.052</td> <td>0.032</td> <td>0.050</td> <td>0.032</td> <td>0.049</td> <td>0.032</td> <td>0.049</td> <td>0.032</td> <td>0.042</td> <td>0.032</td>	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Joint	0.057^{+}	0.032	0.047	0.032	0.047	0.032	0.052	0.032	0.050	0.032	0.049	0.032	0.049	0.032	0.042	0.032	
e -0.011° 0.05 -0.011° 0.002 -0.011° 0.022	-0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 0.014* 0.005	-0011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 -0.011* 0.005 0.011* 0.021	Claims	0.008***	0.001	0.007***	0.001	0.007	0.001	0.008	0.001	0.008***	0.001	0.007	0.001	0.007***	0.001	0.006	0.001	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.012 0.008 0.014 ⁺	0012 0.014 ⁺ 0.008	TeamSize	-0.011^{*}	0.005	-0.011^{*}	0.005	-0.010^{+}	0.005	-0.011 ⁺	0.005	-0.011*	0.005	-0.011^{*}	0.005	-0.011^{*}	0.005	-0.010^{+}	0.005	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.24 ⁴⁺⁺ 0.022 0.230 ⁺⁺ 0.021 0.228 ⁺⁺⁺ 0.021 0.228 ⁺⁺⁺⁺ 0.022 0.24 ⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺⁺	0.244** 0.022 0.230** 0.021 0.244** 0.022 0.241** 0.022 0.241** 0.022 0.240*** 0.022 RO Included Inclu	Family	0.012	0.008	0.014^{+}	0.008	0.014^{+}	0.008	0.010	0.008	0.010	0.008	0.014^{+}	0.008	0.014^{+}	0.008	0.012	0.008	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	RO Included I	R0 Included I	Renewal	0.244***	0.022	0.230***	0.021	0.228	0.021	0.244***	0.022	0.244***	0.022	0.241 ***	0.022	0.240***	0.022	0.232***	0.021	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	beindIncludedIncludedIncludedIncludedIncludedIncludedIncludedJSIncludedIncludedIncludedIncludedIncludedIncludedIncludedJSIncludedIncludedIncludedIncludedIncludedIncludedIncludedLevaron0.039 -0.679° 0.044 -0.510° 0.041 -0.522° 0.042 -0.609° 0.043 -0.717° Included -2612.57 -2550.49 -2607.51 -2605.98 -2569.26 -2569.10 -2574.31 Included -2612.57 -2569.26 -2569.26 -2569.10 -2543.11 Included -2612.57 -2569.26 -2569.26 -2569.10 -2543.11 Squared Correlation 0.382 0.401 0.384 0.384 0.384 0.395 0.405	belowIncludedIncludedIncludedIncludedIncludedIncludedJSIncludedIncludedIncludedIncludedIncludedIncludedAEIncludedIncludedIncludedIncludedIncludedIncludedAE-0.470°0.039-0.679°0.044-0.679°0.041-0.659°0.042-0.696°Aolikelihood-2612.570.039-0.679°0.044-0.6751-2605.98-2669.10Squared Correlation0.3820.4010.3840.3840.395201.75°°0.10.0.3820.4010.3840.3840.3950.3950.3950.10.0.3810.3840.3340.3950.3950.3950.10.0.05.0.3840.3340.3950.3950.3950.10.0.05.0.0510.3840.3050.3950.3950.10.0.05.0.0510.3840.3350.3950.395	Dummy PRO	Included		Included		Included		Included		Included		Included		Included		Included		
Included	JS Included I	JS Included	Dummy period	Included		Included		Included		Included		Included		Included		Included		Included		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	RP Included I	RP Included	Dummy US	Included		Included		Included		Included		Included		Included		Included		Included		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Dummy AEP	Included		Included		Included		Included		Included		Included		Included		Included		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Jolikelihood -2612.57 -2550.48 -2550.49 -2607.51 -2605.98 -2569.26 -2569.10 c 221.94 215.33 215.33 215.33 201.75 201.75 c 221.94 0.401 0.384 0.384 0.384 0.395 0.395	Jolikelihood -2612.57 -2550.48 -2550.49 -2607.51 -2605.98 -2569.16 201,194 2211.94 215.33 215.33 208.45 194.54 215.73 201.75 Squared Correlation 0.382 0.401 0.384 0.384 0.395 0.395 0.10. 0.05. 0.01.	Constant	-0.470	0.039	-0.679****	0.044	-0.679	0.044	-0.510	0.041	-0.552***	0.049	-0.605***	0.042	-0.609****	0.043	-0.717***	0.046	
$221.94^{**} \qquad 215.33^{**} \qquad 215.33^{**} \qquad 208.45^{**} \qquad 194.54^{**} \qquad 215.73^{**} \qquad 201.75^{**} \qquad 0.382 \qquad 0.382 \qquad 0.384 \qquad 0.384 \qquad 0.395 \qquad 0.$: 221.94 215.33 215.33 208.45 194.54 215.33 201.75 201.75 201.75	tic 221.94 215.33 215.33 208.45 194.54 215.33 201.75 201.7	Log Pseudolikelihood	-2612.57		-2550.48		-2550.49		-2607.51		-2605.98		-2569.26		-2569.10		-2543.11		
0.382 0.401 0.401 0.384 0.384 0.395 0.395	Squared Correlation 0.382 0.401 0.401 0.384 0.384 0.395 0.395 0.395	Squared Correlation 0.382 0.401 0.384 0.395 0.395 0.395 0.395 0.305 0.10. 0.10. 0.05.	F statistic	221.94***		215.33***		215.33***		208.45***		194.54***		215.73****		201.75***		191.70***		
	m = 4363	n = 4363. $+ p < 0.10.$ $* p < 0.05.$ $* p < 0.01.$	Multiple Squared Correlation	0.382		0.401		0.401		0.384		0.384		0.395		0.395		0.405		
a = -300. b = 0.10.		*** p < 0.01.	** $p < 0.05$.																	
$^{+}$ p < 0.10. ** p < 0.05.	** $p < 0.05$.		*** $p < 0.01$.																	

Technological Forecasting & Social Change 144 (2019) 25-35

32

understanding of their technology needs. In this way, the environmental and business requirements may be better matched and there will therefore be more chances for governments to supply relevant green energy technologies to the industrial domain, in order to set the basis for their future research activities.

More in detail, we have found that green energy technologies that are more established are more likely to have an impact on the technological development of an industry. Although it may appear that this result suggests policymakers should disregard more basic research, we advocate that PROs should instead focus on both basic and applied research. In other words, PROs should continue refining a technology until it reaches a level that may be understandable and usable by companies, without necessarily limiting their efforts to the creation of a new solution. This may also help to overcome the double externality problem that private research activities have to face by reducing knowledge leakage and uncertainty. Whether a policy can be designed accurately to this aim, is certainly hard to say. However, we retain that technological knowledge generated by PROs (be it more established or less established) can affect the technological development of an industry at two stages (early and mature), and both stages should be stimulated in an appropriate way. On the other hand, sustaining only more established - and mature - technologies, may come at the price of not witnessing a rise in future radical advancements, which may even serve future economic growth by serving as GPTs. Accordingly, the scope of application and technological breadth were both revealed to be core attributes that help make green energy technologies more usable by firms. This implies that green technologies that can readily be applied to diverse domains or which are grounded on a knowledge base that enables several recombinant opportunities will likely spread more easily to multiple companies. Therefore, policymakers are advised that developing more pervasive green solutions or green solutions that are based upon broad innovation activities will sustain the related technological development of an industry.

5.3. Limitations

Some limitations of the current study could not be solved and have been left for future research. First, although the use of patents to capture technology development is widely diffused, especially in the green energy field (Albino et al., 2014; Hoppmann et al., 2013; Popp, 2006), some inventions have not been patented and/or are not patentable. Hence, direct interviews with companies and policymakers would help to close this information gap and would allow the additional characteristics of technologies (e.g., operational complexity) that cannot be captured through patent measures to be scrutinised. Second, in this research, we have only focused on the role of technology attributes, given the limited attention they have received, to analyse the impact of public research. However, we acknowledge that organisational- and environmental-level factors might moderate the technology-level relationships between the attributes of PRO technologies and the technological development of an industry, and this aspect would require multilevel analyses. Third, although the energy sector has been considered particularly relevant for this research, other green domains could also be investigated to ensure the generalisability of the results. Fourth, a properly designed counterfactual impact evaluation could be an important future extension of the work. It could be aimed at testing whether PRO knowledge would have a greater/less impact on the green energy field than knowledge created elsewhere (and by how much). The current design does not allow such a test to be conducted. Finally, from a purely environmental viewpoint, the analysis and its findings do not necessarily imply a positive environmental impact. In other words, it is not known whether the technology attributes under study positively affect the technological development of an industry at the cost of creating technological lock-ins that hamper the uptake of cleaner technologies. However, this kind of investigation is beyond the scope of the current contribution and deserves further ad hoc investigations.

Acknowledgment

The scientific output expressed does not imply a policy position of the European Commission. Neither the European Commission nor any person acting on behalf of the Commission is responsible for the use which might be made of this publication.

References

- Agrawal, A.K., 2001. University-to-industry knowledge transfer: literature review and unanswered questions. Int. J. Manag. Rev. 3, 285–302.
- Ahuja, G., Lampert, C.M., 2001. Entrepreneurship in the large corporation: a longitudinal study of how established firms create breakthrough inventions. Strateg. Manag. J. 22, 521–543.
- Albino, V., Ardito, L., Dangelico, R.M., Messeni Petruzzelli, A., 2014. Understanding the development trends of low-carbon energy technologies: a patent analysis. Appl. Energy 135, 836–854.
- Alkemade, F., Hekkert, M.P., Negro, S.O., 2011. Transition policy and innovation policy: friends or foes? Environ. Innov. Soc. Trans. 1, 125–129.
- Andergassen, R., Nardini, F., Ricottilli, M., 2017. Innovation diffusion, general purpose technologies and economic growth. Struct. Chang. Econ. Dyn. 40, 72–80.
- Appio, F.P., Martini, A., Fantoni, G., 2017. The light and shade of knowledge recombination: insights from a general-purpose technology. Technol. Forecast. Soc. Chang. 125, 154–165.
- Ardito, L., Messeni Petruzzelli, A., Panniello, U., 2016a. Unveiling the breakthrough potential of established technologies: an investigation in the aerospace industry. Tech. Anal. Strat. Manag. 28, 916–934.
- Ardito, L., Messeni Petruzzelli, A., Albino, V., 2016b. Investigating the antecedents of general purpose technologies: a patent perspective in the green energy field. J. Eng. Technol. Manag. 39, 81–100.
- Arts, S., Appio, F.P., Van Looy, B., 2013. Inventions shaping technological trajectories: do existing patent indicators provide a comprehensive picture? Scientometrics 97, 397–419.
- Azagra-Caro, J.M., Barberá-Tomás, D., Edwards-Schachter, M., Tur, E.M., 2017. Dynamic interactions between university-industry knowledge transfer channels: a case study of the most highly cited academic patent. Res. Policy 46, 463–474.
- Balachandra, P., Kristle Nathan, H.S., Reddy, B.S., 2010. Commercialization of sustainable energy technologies. Renew. Energy 35, 1842–1851.
- Banerjee, P.M., Cole, B.M., 2010. Breadth-of-impact frontier: how firm-level decisions and selection environment dynamics generate boundary-spanning inventions. Technovation 30, 411–419.
- Barbieri, N., Ghisetti, C., Gilli, M., Marin, G., Nicolli, F., 2016. A survey of the literature on environmental innovation based on main path analysis. J. Econ. Surv. 30 (3), 596–623.
- Barbieri, N., Marzucchi, A., Rizzo, U., 2018. Knowledge sources and impacts on subsequent inventions: do green technologies differ from non-green ones? In: Science for Policy Research Unit SPRU Working Paper Series N. 11/2018.

Beise, M., Stahl, H., 1999. Public research and industrial innovations in Germany. Res. Policy 28, 397–422.

- Bekkers, R., Bodas Freitas, I.M., 2008. Analysing knowledge transfer channels between universities and industry: to what degree do sectors also matter? Res. Policy 37, 1837–1853.
- Benson, C.L., Magee, C.L., 2014. On improvement rates for renewable energy technologies: solar PV, wind turbines, capacitors, and batteries. Renew. Energy 68, 745–751.
- Bornmann, L., Marx, W., 2014. How should the societal impact of research be generated and measured? A proposal for a simple and practicable approach to allow interdisciplinary comparisons. Scientometrics 98, 211–219.
- Bresnahan, T.F., Trajtenberg, M., 1995. General purpose technologies 'Engines of growth'? J. Econ. 65 (1), 83–108.
- Bruneel, J., D'Este, P., Salter, A., 2010. Investigating the factors that diminish the barriers to university-industry collaboration. Res. Policy 39, 858–868.
- Capaldo, A., Lavie, D., Messeni Petruzzelli, A., 2017. Knowledge maturity and the scientific value of innovations: the roles of knowledge distance and adoption. J. Manag. 43 (2), 503–533.
- Carrillo-Hermosilla, J., Rio, P., Konnola, T., 2010. Diversity of eco-innovations: reflections from selected case studies. J. Clean. Prod. 18 (10), 1073–1083.
- Chen, C.-J., Chang, C.-C., Hung, S.-W., 2011. Influences of technological attributes and environmental factors on technology commercialization. J. Bus. Ethics 104, 525–535.
- Coccia, M., Rolfo, S., 2008. Strategic change of public research units in their scientific activity. Technovation 28, 485–494.
- Cohen, W.M., Levinthal, D.A., 1990. Absorptive capacity: a new perspective on learning and innovation. Adm. Sci. Q. 35, 128–152.
- Cohen, W.M., Nelson, R.R., Walsh, J.P., 2002. Links and impacts: the influence of public research on Industrial R&D. Manag. Sci. 48, 1–23.
- Cohen, J., Cohen, P., West, S.G., Aiken, L.S., 2013. Applied Multiple Regression/ Correlation Analysis for the Behavioral Sciences. Taylor & Francis.
- De Marchi, V., 2012. Environmental innovation and R&D cooperation: empirical evidence from Spanish manufacturing firms. Res. Policy 41 (3), 614–623.
- Dechezleprêtre, A., Martin, R., Mohnen, M., 2014. Knowledge Spillovers from Clean and Dirty Technologies: A Patent Citation Analysis. Working Paper. LSE.
- Dornbusch, F., Neuhäusler, P., 2015. Composition of inventor teams and technological progress – the role of collaboration between academia and industry. Res. Policy 44, 1360–1375.

- Dosi, G., 1982. Technological paradigms and technological trajectories: a suggested interpretation of the determinants and directions of technical change. Res. Policy 11, 147–162.
- Dosi, G., Nelson, R.R., 2010. Chapter 3 technical change and industrial dynamics as evolutionary processes. In: Bronwyn, H.H., Nathan, R. (Eds.), Handbook of the Economics of Innovation. North-Holland, pp. 51–127.
- EC/EACI, 2011. Boosting Green Business. Executive Agency for Competitiveness and Innovation and European Commission, Luxembourg.
- European Environment Agency, 2018. Renewable Energy in Europe 2018 Recent Growth and Knock-on Effects, EEA Report No 20/2018. Publications Office of the European Union, Luxembourg.
- Feldman, M.P., Yoon, J.W., 2011. An empirical test for general purpose technology: an examination of the Cohen–Boyer rDNA technology. Ind. Corp. Chang. 21 (2), 249–275.
- Fischer, T., Leidinger, J., 2014. Testing patent value indicators on directly observed patent value—an empirical analysis of Ocean Tomo patent auctions. Res. Policy 43, 519–529.
- Fleming, L., 2001. Recombinant uncertainty in technological search. Manag. Sci. 47, 117–132.
- Foray, D., Mowery, D.C., Nelson, R.R., 2012. Public R&D and social challenges: what lessons from mission R&D programs? Res. Policy 41, 1697–1702.
- Gambardella, A., Giarratana, M.S., 2013. General technological capabilities, product market fragmentation, and markets for technology. Res. Policy 42, 315–325.
- Gambardella, A., Giuri, P., Luzzi, A., 2007. The market for patents in Europe. Res. Policy 36, 1163–1183.
- Geels, F.W., Kern, F., Fuchs, G., Hinderer, N., Kungl, G., Mylan, J., et al., 2016. The enactment of socio-technical transition pathways: a reformulated typology and a comparative multi-level analysis of the German and UK low-carbon electricity transitions (1990–2014). Res. Policy 45, 896–913.
- Ghisetti, C., Pontoni, F., 2015. Investigating policy and R&D effects on environmental innovation: a meta-analysis. Ecol. Econ. 118, 57–66.

Ghisetti, C., Marzucchi, A., Montresor, S., 2015. The open eco-innovation mode. An empirical investigation of eleven European countries. Res. Policy 44, 1080–1093.

Greene, W., 2004. Fixed effects and bias due to the incidental parameters problem in the Tobit model. Econ. Rev. 23 (2), 125–147.

Griliches, Z., 1987. R&D, Patents and Productivity. University of Chicago Press.

Guan, J.C., Yan, Y., 2016. Technological proximity and recombinative innovation in the alternative energy field. Res. Policy 45, 1460–1473.

- Hargadon, A., 2010. Technology policy and global warming: why new innovation models are needed. Res. Policy 39, 1024–1026.
- Heeley, M.B., Jacobson, R., 2008. The recency of technological inputs and financial performance. Strateg. Manag. J. 29, 723–744.
- Hoppmann, J., Peters, M., Schneider, M., Hoffmann, V.H., 2013. The two faces of market support—how deployment policies affect technological exploration and exploitation in the solar photovoltaic industry. Res. Policy 42, 989–1003.
- Jaffe, A., Palmer, K., 1997. Environmental regulation and innovation: a panel data study. Rev. Econ. Stat. 79 (4), 610–619.
- Jaffe, A.B., Trajtenberg, M., Forgarty, M., 2000. Knowledge spillovers and patent citations: evidence from a survey of inventors. Am. Econ. Rev. 90 (2), 215–218.
- Jaffe, A.B., Newell, R.G., Stavins, R.N., 2005. A tale of two market failures: Technology and environmental policy. Ecol. Econ. 54 (2-3), 164–174.
- Johnstone, N., Haščič, I., Poirier, J., Hemar, M., Michel, C., 2012. Environmental policy stringency and technological innovation: evidence from survey data and patent counts. Appl. Econ. 44 (17), 2157–2170.
- Kaplan, S., Vakili, K., 2015. The double-edged sword of recombination in breakthrough innovation. Strateg. Manag. J. 36, 1435–1457.
- Katila, R., 2002. New product search over time: past ideas in their prime? Acad. Manag. J. 45, 995–1010.
- Köhler, C., Sofka, W., Grimpe, C., 2012. Selective search, sectoral patterns, and the impact on product innovation performance. Res. Policy 41, 1344–1356.
- Lee, Y.-G., Lee, J.-H., 2010. Different characteristics between auctioned and non-auctioned patents. Scientometrics 82, 135–148.
- Lerner, J., 1994. The importance of patent scope: an empirical analysis. RAND J. Econ. 25, 319–333.
- Lindelöf, P., Löfsten, H., 2004. Proximity as a resource base for competitive advantage: university-industry links for technology transfer. J. Technol. Transf. 29, 311–326.
- Long, J.S., 1997. Regression Models for Categorical and Limited Dependent Variables. SAGE Publications.
- Lyall, C., Bruce, A., Firn, J., Firn, M., Tait, J., 2004. Assessing end-use relevance of public sector research organisations. Res. Policy 33, 73–87.
- Maggitti, P.G., Smith, K.G., Katila, R., 2013. The complex search process of invention. Res. Policy 42, 90–100.
- Markard, J., Raven, R., Truffer, B., 2012. Sustainability transitions: an emerging field of research and its prospects. Res. Policy 41 (6), 955–967.
- Martinelli, A., 2012. An emerging paradigm or just another trajectory? Understanding the nature of technological changes using engineering heuristics in the telecommunications switching industry. Res. Policy 41, 414–429.
- Messeni Petruzzelli, A., Rotolo, D., Albino, V., 2015. Determinants of patent citations in biotechnology: an analysis of patent influence across the industrial and organizational boundaries. Technol. Forecast. Soc. Chang. 91, 208–221.
- Mowery, D.C., Nelson, R.R., Martin, B.R., 2010. Technology policy and global warming: why new policy models are needed (or why putting new wine in old bottles won't work). Res. Policy 39, 1011–1023.

Narin, F., Hamilton, K.S., Olivastro, D., 1997. The increasing linkage between U.S. technology and public science. Res. Policy 26, 317–330.

Nemet, G.F., 2012. Inter-technology knowledge spillovers for energy technologies. Energy

Econ. 34, 1259-1270.

- Nerkar, A., Shane, S., 2007. Determinants of invention commercialization: an empirical examination of academically sourced inventions. Strateg. Manag. J. 28, 1155–1166.
- Noailly, J., Shestalova, V., 2017. Knowledge spillovers from renewable energy technologies: lessons from patent citations. Environ. Innov. Soc. Trans. 22 (March), 1–14.
- Nooteboom, B., Vanhaverbeke, W., Duysters, G., Gilsing, V., van den Oord, A., 2007. Optimal cognitive distance and absorptive capacity. Res. Policy 36, 1016–1034.
- Novelli, E., 2015. An examination of the antecedents and implications of patent scope. Res. Policy 44, 493–507.
- OECD, 2012. OECD Green Growth Studies Energy. OECD Publishing.
- OECD/IEA, 2015. Energy and Climate Change, World Energy Outlook Special Report. IEA Publishing, Paris.
- Oltra, V., Saint Jean, M., 2009. Sectoral systems of environmental innovation: an application to the French automotive industry. Technol. Forecast. Soc. Chang. 76 (4), 567–583.
- Oltra, C., Upham, P., Riesch, H., Boso, A., Brunsting, S., Dütschke, E., Lis, A., 2012. Public responses to CO2 storage sites: lessons from five European cases. Energy Environ. 23, 227–248.
- Popp, D., 2006. Innovation in climate policy models: implementing lessons from the economics of R&D. Energy Econ. 28, 596–609.
- Popp, D., 2017. From science to technology: the value of knowledge from different energy research institutions. Res. Policy 46 (9), 1580–1594.
- Popp, D., Newell, R., 2012. Where does energy R&D come from? Examining crowding out from energy R&D. Energy Econ. 34 (4), 980–991.

Popp, D., Newell, R.G., Jaffe, A.B., 2010. Chapter 21 - energy, the environment, and technological change. In: Hall, B.H., Rosenberg, N. (Eds.), Handbook of the Economics of Innovation: North-Holland. vol. 2. pp. 873–937.

Reitzig, M., 2004. Improving patent valuations for management purposes—validating new indicators by analyzing application rationales. Res. Policy 33 (6), 939–957.

Rennings, K., 2000. Redefining innovation — eco-innovation research and the contribution from ecological economics. Ecol. Econ. 32, 319–332.

Rennings, K., Rammer, C., 2009. Increasing energy and resource efficiency through innovation – an explorative analysis using innovation survey data. Czech J. Econ. Finan. 59 (1), 442–459.

Rosenberg, N., Trajtenberg, M., 2004. A general-purpose technology at work: the Corliss steam engine in the late-nineteenth-century United States. J. Econ. Hist. 64, 61–99.

Salter, A.J., Martin, B.R., 2001. The economic benefits of publicly funded basic research: a critical review. Res. Policy 30, 509–532.

- Savino, T., Messeni Petruzzelli, A., Albino, V., 2015. Search and recombination process to innovate: a review of the empirical evidence and a research agenda. Int. J. Manag. Rev. 19 (11), 54–75.
- Shane, S., 2001. Technological opportunities and new firm creation. Manag. Sci. 47, 205–220.
- Singh, J., 2008. Distributed R&D, cross-regional knowledge integration and quality of innovative output. Res. Policy 37 (1), 77–96.

Sohn, S.Y., Lee, W.S., Ju, Y.H., 2013. Valuing academic patents and intellectual properties: different perspectives of willingness to pay and sell. Technovation 33, 13–24.

- Sterzi, V., 2013. Patent quality and ownership: an analysis of UK faculty patenting. Res. Policy 42, 564–576.
- Story, V.M., Daniels, K., Zolkiewski, J., Dainty, A.R.J., 2014. The barriers and consequences of radical innovations: introduction to the issue. Ind. Mark. Manag. 43, 1271–1277.
- Teece, D., Pisano, G., Shuen, A., 1997. Dynamic capabilities and strategic manage-ment. Strateg. Manag. J. 18 (7), 509–533.
- Trajtenberg, M., Henderson, R., Jaffe, A., 1997. University versus corporate patents: a window on the basicness of invention. Econ. Innov. New Technol. 5, 19–50.
- Unruh, G.C., 2000. Understanding carbon lock-in. Energy Policy 28 (12), 817–830. Veugelers, R., 2012. Which policy instruments to induce clean innovating? Res. Policy 41,

1770–1778. Wainstein, M.E., Bumpus, A.G., 2016. Business models as drivers of the low carbon power

system transition: a multi-level perspective. J. Clean. Prod. 126, 572–585. Wiersema, M.F., Bowen, H.P., 2009. The use of limited dependent variable techniques in

strategy research: issues and methods. Strateg. Manag. J. 30, 679–692.

 WIPO, 2012. IPC Green Inventory. WIPO Publication No. L434/9(E). 978-92-805-2113-9.
 Wooldridge, J., 2012. Introductory Econometrics: A Modern Approach. Cengage Learning.

Ziedonis, A.A., 2007. Real options in technology licensing. Manag. Sci. 53, 1618–1633.

Lorenzo Ardito, PhD, is Assistant Professor at the Campus "Bio-Medico" University of Rome, Italy, and a former Postdoctoral Research Fellow in Technology and Innovation Management at Politecnico di Bari, Italy. He has been a visiting PhD candidate at WHU-Otto Beisheim School of Management, Germany. His main research interests concern the analysis of the dynamics underlying the development, commercialization, and societal impacts of new technologies. He has published in leading international journals as Technological Forecasting and Social Change, R&D Management, Journal of Engineering and Technology Management, Corporate Social Responsibility and Environmental Management, European Management Journal, European Management Review, Technology Analysis and Strategic Management, and Applied Energy. He serves as Guest Editor for several journals. Finally, he has been awarded as Outstanding Reviewer for Journal of Knowledge Management in the Emerald Literati Network 2016 Awards for Excellence.

Antonio Messeni Petruzzelli, PhD, is Professor in Innovation Management and cofounder of the Innovation-Management Group at the Politecnico di Bari. He is the author of more than 50 international publications and three international books. Specifically, his research lies at the intersection between innovation and knowledge management. His studies on this issue have been published in leading journals such as Academy of Management Perspectives, Journal of Management, International Journal of Management Reviews, Long Range Planning, Technological Forecasting and Social Change, and Journal of Organizational Behavior. He belongs to the editorial team of Journal of Knowledge Management and serves as guest-editor for journals such as Technological Forecasting and Technovation.

Claudia Ghisetti, PhD, is Assistant Professor at the Catholic University of Milan, Department of Economic Policy. Former researcher at the Joint Research Centre of the European Commission. She holds a PhD in European Law and Economics (University of Bologna) and she has been visiting researcher at GREDEG-CNRS, University of Nice-Sophia Antipolis and at the Center for European Economic Research (ZEW) in Mannheim. Before joining the Catholic University she has been post-doctoral research fellow at the Department of Economics and Management of the University of Ferrara (IT), and Research Associate at the inter-university Research Center on Sustainability Environmental Economics and Dynamics Studies (SEEDS). Her expertise is in microeconomics empirical analysis in the fields of economics of innovation and environmental economics. She published her research in international journals such as Ecological Economics, Research Policy, Technological forecasting and Social Change, Climate Policy, the Journal of Economic Surveys and the Journal of Cleaner Production.